Author’s response to reviews

Title: Contraceptive use intentions and unmet need for family planning among reproductive-aged women in the Upper East region of Ghana

Authors:
Ayaga A. Bawah (aabawah@ug.edu.gh;aabawah@gmail.com)
Patrick O. Asuming (poa2102@gmail.com)
Sebastian F. Achana (fsachana@gmail.com)
Edmund Wedam Kanmiki (eddy100@hotmail.com)
John Koku Awoonor-Williams (kawoonor@gmail.com)
James F. Phillips (jfp2113@cumc.columbia.edu)

Version: 1  Date: 30 Apr 2018

Author’s response to reviews:

Dear Editor,

We have revised the manuscript to include an explicit statement about the informed consent process and are resubmitting the revised manuscript for review.

Thank you.